Author's response to reviews

Title: Implementing chlamydia screening: what do women think? A systematic review of the literature

Authors:

Natasha L Pavlin (npavlin@ozisp.com.au)
Jane M Gunn (j.gunn@unimelb.edu.au)
Rhian Parker (r.parker@unimelb.edu.au)
Christopher K Fairley (cfairley@unimelb.edu.au)
Jane Hocking (j.hocking@unimelb.edu.au)

Version: 6  Date: 1 September 2006

Author's response to reviews:

Changes have been made as suggested by the editors - the two tables are now included in the body of the article and the additional files will be removed. The references that were in superscript are now in normal text. The paper remains otherwise unchanged.